Author's response to reviews

Title: Mammography stages of change in middle-aged women with schizophrenia: An exploratory analysis

Authors:

Laurie A Lindamer (llindamer@ucsd.edu)
Emily Wear (emngreenpastures@yahoo.com)
Georgia Robins Sadler (gsadler@ucsd.edu)

Version: 2 Date: 24 October 2006

Author's response to reviews:

We have made the requested Formatting changes. Please see the detailed responses below.

1. Please remove the 'For:', 'Word count' and 'Tables' text from the title page.
Done.

2. Qualifications - Please remove the authors' qualifications (e.g. PhD / MD) from the title page.
Done.

3. Affiliations - Each affiliation should be written in full, this may mean repeating institutional addresses if the authors are from different departments within an institution, and be provided in the following format: department, institute, city, country. Can we also ask you to link each author to their corresponding affiliation with the use of superscript numbers. Please do not include full stops at the end of the addresses.
Done.

4. Keywords section - Please remove the keywords section from the manuscript.
Done.

5. Introduction - Please rename this section 'Background'.
Done.

6. Figure and table placement text - Please remove this.
Done.